\begin{table}%t1
 %\centering
\par
    \caption{Adopted conversion relations for extinction values, transforming $\EBV \ {\rm (X)}$ to
    the \citet{Laney2007} system, by $\EBV \ {\rm (L07)} = a \ \EBV \ {\rm (X)} + b$.}
    \begin{tabular}{llcccrc}
      \hline\hline %\\
      Authors & F95 id & slope & intercept & $\sigma$ & $N$ & weight \\
      \hline %\\
      \citet{Parsons1975}       & PB   & $0.908 \pm 0.026$ &                    & 0.064 & 105 & 0.3 \\
      \citet{Yakimova1975}      & YT   & $0.863 \pm 0.017$ & $0.036 \pm 0.007$  & 0.034 & 82  & 1.2 \\
      \citet{Janot-Pacheco1976} & JP   & $0.873 \pm 0.077$ &                    & 0.089 & 30  & 0.1 \\
      \citet{Dean1978}          & DWC  & $0.961 \pm 0.012$ & $-0.014 \pm 0.005$ & 0.025 & 77  & 3.2 \\
      \citet{Pel1978}           & PE   & $0.945 \pm 0.029$ &                    & 0.047 & 64  & 0.5 \\
      \citet{Kron1979}          & KR   & $0.867 \pm 0.028$ &                    & 0.072 & 124 & 0.2 \\
      \citet{Feltz1980}         & FM   & $0.890 \pm 0.022$ & $0.039 \pm 0.013$  & 0.046 & 54  & 0.6 \\
      \citet{Dean1981}          & DE   & $0.857 \pm 0.034$ &                    & 0.038 & 35  & 0.9 \\
      \citet{Turner1987}        & TLE  & $0.938 \pm 0.050$ &                    & 0.037 & 21  & 1.0 \\
      \citet{Schechter1992}     & SACK & $0.861 \pm 0.073$ & $0.039 \pm 0.037$  & 0.034 & 9   & 1.2 \\
      \citet{Laney1993}         & LS   & $0.985 \pm 0.008$ &                    & 0.015 & 45  & -   \\
      \citet{Fernie1995}        & FE   & $0.952 \pm 0.010$ &                    & 0.029 & 147 & 1.9 \\
      \citet{Eggen1996}         & EG2  & $0.754 \pm 0.045$ & $0.064 \pm 0.013$  & 0.043 & 34  & 0.7 \\
      \citet{Bersier2002}       &      & $0.883 \pm 0.021$ & $0.035 \pm 0.009$  & 0.040 & 64  & 0.8 \\
      \hline
    \end{tabular}
    \label{tab:f95l07}
 \end{table}